Author’s response to reviews

Title: Design and Implementation of a Cross-sectional Nutritional Phenotyping Study in Healthy US Adults

Authors:

Lacey Baldiviez (lmbaldiviez@ucdavis.edu)
Nancy Keim (nancy.keim@ars.usda.gov)
Kevin Laugero (kevin.laugero@ars.usda.gov)
Daniel Hwang (daniel.hwang@ars.usda.gov)
Liping Huang (liping.huang@ars.usda.gov)
Leslie Woodhouse (leslie.woodhouse@ars.usda.gov)
Dustin Burnett (dustin.burnett@ars.usda.gov)
Melissa Zerofsky (mzerofsky@gmail.com)
Ellen Bonnel (ellen.bonnel@ars.usda.gov)
Lindsay Allen (lindsay.allen@ars.usda.gov)
John Newman (john.newman@ars.usda.gov)
Charles B. Stephensen (charles.stephensen@ars.usda.gov)

Version: 1 Date: 18 Sep 2017

Author’s response to reviews:

I have made the suggested changes, regarding the figure, and have adjusted formatting of the text according to journal policy.